

Impact of searching clinical trial registries in systematic reviews of pharmaceutical treatments: methodological systematic review and reanalysis of meta-analyses

Marie Baudard,^{1,2} Amélie Yavchitz,^{1,2,3} Philippe Ravaud,^{1,2,3,4,5} Elodie Perrodeau,^{1,2,3,4} Isabelle Boutron^{1,2,3,4}

¹Université Paris Descartes, Sorbonne Paris Cité, Paris, France

²Centre de Recherche Epidémiologie et Statistique, INSERM U1153, Paris, France ³Cochrane France, Paris, France

⁴Centre d'Épidémiologie Clinique, Hôpital Hôtel-Dieu, Assistance Publique-Hôpitaux de Paris, 75004 Paris, France

⁵Department of Epidemiology, Mailman School of Public Health, Columbia University, New York, NY, USA

Correspondence to: A Yavchitz amelie.yavchitz@aphp.fr Additional material is published

online only. To view please visit the journal online (http://dx.doi. org/10.1136/bmj.j448)

Cite this as: *BMJ* **2017;356:j448** http://dx.doi.org/10.1136/bmj.j448

Accepted: 18 January 2017

ABSTRACT

OBJECTIVE

To evaluate the impact of searching clinical trial registries in systematic reviews.

DESIGN

Methodological systematic review and reanalyses of meta-analyses.

DATA SOURCES

Medline was searched to identify systematic reviews of randomised controlled trials (RCTs) assessing pharmaceutical treatments published between June 2014 and January 2015. For all systematic reviews that did not report a trial registry search but reported the information to perform it, the World Health Organization International Trials Registry Platform (WHO ICTRP search portal) was searched for completed or terminated RCTs not originally included in the systematic review.

DATA EXTRACTION

For each systematic review, two researchers independently extracted the outcomes analysed, the number of patients included, and the treatment effect estimated. For each RCT identified, two researchers independently determined whether the results were available (ie, posted, published, or available on the sponsor website) and extracted the data. When additional data were retrieved, we reanalysed meta-analyses and calculated the weight of the additional RCTs and the change in summary statistics by comparison with the original meta-analysis.

RESULTS

Among 223 selected systematic reviews, 116 (52%) did not report a search of trial registries; 21 of these did not report the information to perform the search (key words, search date). A search was performed for 95 systematic reviews; for 54 (57%), no additional RCTs were found and for 41 (43%) 122 additional RCTs were identified. The search allowed for increasing the number of patients by more than 10% in 19 systematic reviews, 20% in 10, 30% in seven, and 50% in four. Moreover, 63 RCTs had results available; the results for 45 could be included in a meta-analysis. 14 systematic

WHAT IS ALREADY KNOWN ON THIS TOPIC

Despite recommendation, searches of clinical trial registries are not systematically reported by authors of systematic reviews

WHAT THIS STUDY ADDS

Searching clinical trial registries does identify additional trials for a systematic review of the literature, thus increasing the potential value of the review

reviews including 45 RCTs were reanalysed. The weight of the additional RCTs in the recalculated metaanalyses ranged from 0% to 58% and was greater than 10% in five of 14 systematic reviews, 20% in three, and 50% in one. The change in summary statistics ranged from 0% to 29% and was greater than 10% for five of 14 systematic reviews and greater than 20% for two. However, none of the changes to summary effect estimates led to a qualitative change in the interpretation of the results once the new trials were added.

CONCLUSIONS

Trial registries are an important source for identifying additional RCTs. The additional number of RCTs and patients included if a search were performed varied across systematic reviews.

Introduction

Systematic reviews are considered to provide the highest level of evidence.12 They are widely used by the developers of clinical practice guidelines, granting health agencies, and journal editors.³⁻⁶ A major challenge of systematic reviews is to identify all relevant randomised controlled trials (RCTs), whatever their publication status.7-13 Indeed, results for half of RCTs are never published and the publication status is affected by the nature and direction of results, which may bias the results of the systematic review.¹⁴ In some cases, the importance of unpublished trials can be considerable-for example, the addition of unpublished data in the updated Cochrane review assessing the efficacy of neuraminidase inhibitors for influenza modified the conclusion.¹⁵¹⁶ Initiatives aimed at reducing publication bias include the trial registration policy initiated by the International Committee of Medical Journal Editors (ICMJE) in 2005.17 In 2007, the US Food and Drug Administration Amendments Act required the posting of clinical trial results at ClinicalTrials.gov no later than one year after the date of final collection of data for the prespecified primary outcome, for all phase II to IV trials of drugs, biological treatments, and devices.¹⁸¹⁹ The research community has embraced this policy, and there was a noticeable increase in trial registration around the time of implementation of the ICMJE policy.20 In April 2016, about 90000 completed experimental studies were registered at ClinicalTrials. gov (the largest registry), and 16500 have results posted.

When performing systematic reviews, the search of trial registries is now considered an essential tool.^{3 21-23} Previous studies showed that searches of clinical trial registries are not systematically reported by authors of

systematic reviews,²⁴⁻²⁶ but to our knowledge, none had systematically performed a search of trial registries to quantify the impact of searches (ie, to quantify the number of missing trials identified by a search and the change in summary statistics when these missing trials are considered). In this study we describe whether and how clinical trial registries were searched in published systematic reviews of pharmaceutical treatments and evaluate the impact of searching registries on the identification of additional RCTs (ie, eligible completed or terminated RCTs not included in the systematic review).

Methods

Identification of systematic reviews Search strategy

On 16 March 2015 we systematically searched Medline through PubMed for all systematic reviews of RCTs assessing pharmaceutical treatments that were published in English between 1 June 2014 and 31 January 2015. We did this by searching for "Meta-Analysis[ptyp] AND ("2014/06/01"[PDAT]: "2015/01/31"[PDAT]) AND

English[lang] appearing in the title, abstract, or keywords.

Inclusion and exclusion criteria

One researcher screened the titles and abstracts of citations retrieved to identify all reports of systematic reviews of RCTs with at least one meta-analysis including at least two RCTs and assessing a pharmaceutical treatment (ie, drug, health related biological product, or nutritional supplement). We excluded updates of previously published systematic reviews and systematic reviews of diagnostic test accuracy, prognosis, economics evaluations, genetics, non-RCT studies, network meta-analyses, and indirect comparison meta-analysis as well as individual patient data meta-analyses. The full text of potentially relevant citations was obtained. As a quality control procedure, another researcher independently screened 20% of the citations and confirmed the eligibility of all systematic reviews included. Discrepancies were discussed to reach consensus.

Data extraction

From the published reports and supplementary appendices, two researchers independently recorded, when available, the general characteristics of the systematic review (type of journal: general medical, specialty, or Cochrane review), the funding source (not-for-profit, for-profit, not reported, or unclear), and the number of RCTs and participants included in the systematic review. They also recorded the reporting of the clinical trial registry search (ie, whether such a search was reported, the name and type of registries searched, and whether the results of the search (the number and identification of RCTs identified from the search) were reported. Any disagreement was resolved by discussion and consensus.

Impact of searching clinical trial registries

For each systematic review that did not report a search in clinical trial registries, we systematically performed a search reproducing the conditions of the original search reported in the systematic review, particularly taking into account the date of the search and the inclusion criteria of the systematic review.

Search strategy

Our search strategy followed the same search and selection process described by the authors of the published systematic reviews.

Firstly, from the selected full text articles and all available supplementary materials we systematically recorded the search terms related to the condition and interventions used by authors and the date of the last electronic search. From this analysis we excluded systematic reviews that did not provide search terms or the date of the search.

Secondly, we searched the World Health Organization International Clinical Trials Registry Platform (WHO ICTRP) search portal, which contains the trial registration datasets provided by several registries. We chose this portal because it includes 16 national and international primary registries, including ClinicalTrials.gov. In the advanced search window of the portal (http://apps.who. int/trialsearch/) we entered the search terms recorded in the "condition" and "intervention" fields with Boolean operators. We chose "all" in the "recruitment status" field and "Search for clinical trials in children" when appropriate. Appendix 1 provides details of the search strategies and keywords for each systematic review.

Identification of completed or terminated RCTs

For each search we downloaded all the citations retrieved and identified all studies with a recruitment status recorded as "completed" or "terminated".

For each systematic review, two researchers independently screened the records retrieved and selected all completed or terminated RCTs not already included in the systematic review that fulfilled the systematic review's eligibility criteria for participants, interventions, and comparator. We systematically verified in the history or archives of the registry that the recruitment status was recorded as completed or terminated before the date of the search (see appendix 2). Any disagreements were resolved by consensus. A third researcher screened all selected records to confirm their inclusion.

Availability of RCT results

Two researchers independently determined whether the trial results were available (posted, published, or available on the sponsor's website) for each selected RCT. We searched for results posted on clinical trial registries and publications referenced on the trial registry, performed an electronic search of PubMed and Google, and searched the sponsor's website. All trials with results available were screened, and we selected only trials for which the results became available before the last electronic search of the systematic review.

Inclusion of the RCT results in meta-analyses We recorded the number of meta-analyses reported in

the systematic review, the number of meta-analyses

that could include the additional RCTs, and the number of meta-analyses for which all the RCTs identified had results available and could be included in the meta-analysis.

Finally, we determined the impact of including the RCTs on treatment effect estimates. For this purpose, we used an algorithm to select one meta-analysis in which at least one RCT with results available could be included.

For each systematic review we recorded all the outcomes of the meta-analyses reported in the systematic review report. For each eligible RCT with results available, we determined whether the RCT could be included in the meta-analyses previously recorded—ie, the RCT reports included:

Continuous outcomes—sample size, mean, and one measure of dispersion (standard deviation, standard error, or confidence interval) by group. Standard errors and confidence intervals were converted into standard deviations to perform the meta-analyses. When results were given for separate subgroups, we pooled the results, the pooled sample size being the sum of the subgroup sample sizes and the pooled mean being the weighted mean of the subgroups and the pooled standard deviation combined.³

Binary outcomes—sample size and number of events by group.

Time-to-event outcomes—hazard ratio and 95% confidence interval or median survival times and confidence intervals by group.²⁷

Finally, when RCTs could be included in several meta-analyses, we selected only one meta-analysis according to the following order of outcomes analysed: the primary efficacy outcome of the systematic review, the primary harms outcome, and the patient important outcome, such as mortality, quality of life, or morbidity. If several of these outcomes could be used to include new RCTs, we selected the first meta-analysis reported. If none of these outcomes could be used to include a new RCT, we selected the first meta-analysis reported.

For each meta-analysis selected, we extracted the outcome data from the RCTs identified (eg, number of events and number of patients in each group, means, standard deviations). When the outcome data were available from several sources, we extracted a single source according to a prespecified order; data reported in the registry, data reported in a published report, and data reported on the sponsor's website.

Data analysis

R v3.1.0 (www.r-project.org), the R Project for Statistical Computing, was used for statistical analysis. Qualitative variables are represented by percentages, and quantitative variables by medians (interquartile ranges). In a post hoc analysis, we used a χ^2 test to compare the proportion of reviews reporting a trial registry search according to the type of systematic review (Cochrane *v* non-Cochrane) and funding source (not-for-profit or not funded *v* for-profit, funding not reported or unclear).

For the meta-analysis selected for recalculation (one for each selected systematic review), we calculated the

summary statistics (risk ratios, odds ratios, hazard ratios, mean differences, or standardised mean differences) and the I² statistic (measure of heterogeneity) with and without trials retrieved by a trial registry search. We used similar methods to those of Hart et al.²⁸ We reported the magnitude of the change in the result of the meta-analysis as a percentage change in the summary statistic after including data from the RCTs retrieved after the registry search. For risk ratios and odds ratios, we calculated the percentage change of the log transformation as $(\log(E) - \log(I)) \times 100 / \log(E)$, where E is the effect estimate excluding newly retrieved data and I is the effect estimate including newly retrieved data. We calculated the log transformation for relative risks and odds ratios so that the point of "no effect" was equal to 0 instead of 1, thus allowing for a calculation of percentage change. For weighted mean differences, we calculated the percentage change by using the formula (E-I)×100/E.

We reanalysed the published meta-analyses by using the same statistical method (Peto, Mantel-Haenszel, inverse variance), analysis model (fixed ν random effects), and measure of effect (risk ratio, odds ratio, weighted mean difference) used by the original authors. For all meta-analyses, we assessed heterogeneity by calculating the I² statistic and τ^2 (DerSimonian-Laird estimate).

Results

Identification and characteristics of reports

Among the 2249 citations retrieved, we included 223 reports of systematic reviews with meta-analyses (fig 1). Table 1 lists the characteristics of the included systematic reviews. One third (35%) were Cochrane reviews; a median 10 (interquartile range 6-18) RCTs were included in the systematic reviews, ⁶⁻¹⁸ with a median 1594 (614-5027) patients.

Reporting of registry search in systematic reviews

Among the 223 systematic review reports included, 107 (48%) mentioned searching at least one clinical trial registry: 48 of these (45%) reported searching only individual registries, 11 (10%) only portals, and 44 (41%) a combination of individual registries and portals. Four did not report the type of registry or portal searched. The portal and individual register most frequently searched were the WHO ICTRP search portal (n=53, 50%) and ClinicalTrials. gov (n=89, 83%), and for 40 studies (37%), both were searched. In only 47 of the 107 (21%) reports were the results of the clinical trial registry search clearly described (ie, with a description of the number and identification of RCTs found from the search) (fig 1, table 1): 16 of these 47 reviews (34%) did not retrieve any eligible RCTs, 11 (23%) retrieved only ongoing studies, 13 (28%) retrieved at least one completed or terminated RCT with no results available, and 7 (15%) retrieved at least one completed or terminated RCT with results identified. Of these last seven, three included RCTs in at least one meta-analysis.

A search of a trial registry was more frequent in Cochrane than in non-Cochrane reviews (65/77 (84%) v 42/146 (29%), P<0.001) and in not-for-profit funding or



Fig 1 | Study flow diagram

no funding than for-profit funding, funding not reported, or funding unclear (79/139 (57%) ν 28/84 (33%), P<0.001).

Impact of searching registries

Identification of completed or terminated RCTs

Among the 116 systematic reviews not reporting a search in trial registries, we were not able to perform the registry search for 21 (18%) because the search date or the keywords were not reported. Therefore the search was performed for 95 systematic reviews. Among the 15282 records screened (median 23 (interguartile range 6-150) for each systematic review), we identified 122 eligible RCTs terminated or completed (involving 52743 patients) not originally included in the systematic review. Among the 122 RCTs, 104 (85%) were classified as completed and 18 (15%) as terminated. Among the 18 RCTs classified as terminated, three had results available and were included in meta-analyses: two were stopped early because of adverse events and one was stopped early because of futility. The remaining 15 RCTs had no results available and no information on the reason for stopping early.

Availability of RCT results

Overall, *t*he trial registry searches identified at least one eligible RCT for 41 of 95 (43%) systematic reviews, with a median 9% (interquartile range 4 to 18)⁴⁻¹⁸ additional patients for each systematic review (fig 2, table 2, see

appendix 3). Among these 41 systematic reviews with additional RCTs identified, the number of patients included increased by 10% to 20% in nine, 20% to 30% in three, 30% to 40% in two, 40% to 50% in one, and more than 50% in four.

We identified results for 63 of 122 RCTs (52%) involving 42 202 patients. Of these 63 RCTs, 45 (71%) involving 21 358 patients could be included in the quantitative analysis (ie, reported sufficient data to be included in at least one meta-analysis of the systematic review). The 18 remaining RCTs with results could not contribute to the quantitative analysis because of differences in definition or metrics used between the outcome reported in the RCT and the outcome of the systematic review or outcome reporting bias.

The results of the 63 RCTs identified were posted (n=41, 65%), published as identified by a reference reported on the registry (n=21, 33%) or from a complementary search (n=10, 16%), or were available on the company's website (n=31, 49%). The results were available in one (n=29, 46%), two (n=27, 43%), or three sources (n=7, 11%).

For 14 systematic reviews, the trial registry searches allowed for identifying RCTs with results (n=45) that could contribute to the quantitative analysis. Among the 73 meta-analyses reported in these 14 systematic reviews; the search in trial registries retrieved additional results that could be included in 59 meta-analyses. Overall, 31 of 59 (53%) meta-analyses were considered complete (ie, all the RCTs identified had available results and could be included in the meta-analysis).

Inclusion of RCT results in meta-analyses

Finally, we recalculated the effect estimates for the selected meta-analyses from the 14 systematic reviews including RCTs that could contribute to the quantitative analysis. The 14 meta-analyses selected included eight efficacy outcomes and six harms outcomes. In the meta-analysis without additional RCTs, results for 12 of 14 outcomes statistically significantly favoured the experimental treatment, and results for two did not differ from the comparator.

The weight of the eligible RCTs included ranged from 0.2% to 58% and was greater than 10% for five of 14 systematic reviews, 20% for three, and 50% for one. The change in summary statistics ranged from 0% to 29% and was greater than 10% for five of 14 systematic reviews and greater than 20% for two. For example, in the meta-analysis with a 29% change in summary effect, the mean difference changed from -0.35 (95% confidence interval -0.51 to -0.19) to -0.45 (-0.55 to -0.36), for a larger effect after inclusion of the new RCTs. However, including the RCTs identified by a trial registry search did not change the statistical significance or direction of the 14 meta-analyses.

Discussion

Despite recommendations,²³ about half of the published systematic reviews performed a search of clinical

Table 1 | Characteristics of included systematic reviews and registry searches. Values are numbers (percentages) unless stated otherwise

| Characteristics | Systematic reviews (n=223) |
|--|-------------------------------|
| Characteristics of reviews | |
| Type of review: | |
| Cochrane | 77 (35) |
| Non-Cochrane | 146 (65) |
| Funding: | |
| Not-for-profit | 106 (47) |
| For-profit | 3 (1) |
| No funding | 33 (15) |
| Not reported or unclear | 81 (36) |
| No of RCTs included: | |
| Median (interquartile range) | 10 (6-18) |
| Minimum-maximum | 2-158 |
| No of patients included*: | |
| Median (interquartile range) | 1594 (614-5027) |
| Minimum-maximum | 47-102607 |
| Clinical trial registry search | 107 (48.0) |
| Characteristics of registry search (n=107) | |
| Search portal (at least one portal searched): | 57 (53) |
| WHO ICTRP | 53 (49) |
| metaRegister of Controlled Trials | 15 (14) |
| International Federation of Pharmaceutical Manufacturers & Associations | 1 (1) |
| Individual clinical trial registries approved by WHO or ICMJE (at least one searched): | 93 (87) |
| ClinicalTrials.gov | 89 (83) |
| ISRCTN Registry | 22 (21) |
| EU Clinical Trials Register | 5 (5) |
| Australian New Zealand Clinical Trials Registry | 5 (5) |
| Japan Primary Registries Network | 3 (3) |
| Chinese Clinical Trial Registry | 1 (1) |
| Non-approved or unclear individual clinical trial registries | 11 (10) |
| RCTs=randomised controlled trials: WHO ICTRP=World Health Organization International Trial | sRegistry |

RCTs=randomised controlled trials; WHO ICTRP=World Health Organization International Trials Registry Platform; ICMJE=International Committee of Medical Journal Editors; ISRCTN=International Standard Randomised Controlled Trial.

*Number unclear or missing in nine non-Cochrane systematic reviews.



Fig 2 | Identification of trials by searching clinical trial registries

trial registries and only one fifth reported the results of the search. When we performed the registry search, we identified additional studies for 43% of the systematic reviews. We reanalysed 14 meta-analyses to include data from randomised controlled trials (RCTs) retrieved by the trial registry search. The addition of data from registries mainly adds to the precision of summary estimates, but none of the changes led to a qualitative change in the interpretation of the results once the new trials were added.

Comparison with other studies

Our results are consistent with other studies, showing that the search for unpublished trial data is still often lacking in systematic reviews.^{24-26 29 30} In a random sample of 300 recent systematic reviews indexed in Medline in February 2014, only 19% reported searching trial registries.³¹ A previous study by Hart in 2012 aimed to reanalyse meta-analyses by adding unpublished outcome data from trials obtained from the US Food and Drug Administration to published meta-analyses.²⁸ The study documented that the addition of unpublished data obtained from the FDA could change the magnitude of the effect size or, in a few cases, the statistical significance of meta-analyses. A systematic review in 2016, aimed at quantifying the impact of the underreporting of adverse events in systematic reviews, showed that the inclusion of unpublished data might reduce the imprecision of pooled effect estimates in meta-analysis of adverse events.32 However, to our knowledge the impact of searching trial registries in terms of identifying trials and their inclusion in the analysis when results are available has never been evaluated.

Limitations of this study

Our study has some limitations. Firstly, we searched only the World Health Organization International Trials Registry Platform (WHO ICTRP) search portal using the keywords reported by authors for their electronic search. Consequently, we cannot claim that we identified all RCTs. However, this portal brings together 16 national and international primary registries, including Clinical-Trials.gov. Furthermore, in a previous study, the overlap between ICTRP and ClinicalTrials.gov was good, because all records identified in ClinicalTrials.gov were also identified in WHO ICTRP.33 Secondly, we did not account for eligibility criteria related to trial quality. The quality assessment of data recorded from trial registries is difficult and some trials could secondarily be excluded because of insufficient quality. Thirdly, we did not attempt to contact investigators of the unpublished trials to obtain results. In fact, we aimed to reproduce the condition the authors encountered and it would not be appropriate to ask authors for results after such a delay. Furthermore, we did not search for additional data presented in conference abstracts or search FDA and European Medicines Agency websites. Therefore, the number of systematic reviews with trials identified by a search of clinical trial registries and the amount of data from RCTs retrieved from such registries might be underestimated. Fourthly, to make the workload manageable, we

| | No of RCTs (| No of patients | (5 | | Source of summary statisti | | | | | |
|---|--|--|---|--|---|--|---|--|---|----------------------------------|
| Outcome and trial ID | s In original SR | Retrieved from search | Retrieved with results that could contribute to ≥MA | Jescription of selected outcomes (type) | Selected MA in original SR | Selected MA with new RCTs included | Weight of new RCTs included in selected MA (%) | Change in summary statistic (%) | Direction of change in summary statistic | Change statistic significa |
| Efficacy: | | | | | | | | | | |
| - | 18 (9952) | 1 (73) | 1 (73) | Atrial fibrillation (PO) | OR 0.51 (0.36 to 0.70) | OR 0.53 (0.38 to 0.73) | 1.9 | 6 | Decrease efficacy | No |
| 2 | 9 (11 390) | 2 (355) | 1 (322) | PASI 75 (PO) | RR 18.28 (12.76 to 26.17) | RR 14.20 (10.72 to 18.81) | 37.6 | 6 | Decrease efficacy | No |
| m | 20 (8225) | 8 (1806) | 2 (1400) | Overall survival (U) | HR 0.87 (0.82 to 0.91) | HR 0.88 (0.84 to 0.93) | 14.8 | 8 | Decrease efficacy | No |
| 4 | 6 (2264) | 1 (1029) | 1 (1029) | Overall survival (U) | HR 0.89 (0.80 to 0.99) | HR 0.90 (0.83 to 0.98) | 34.8 | 10 | Decrease efficacy | No |
| 5 | 12 (6297) | 2 (340) | 1 (102) | Overall survival (U) | HR 0.99 (0.90 to 1.09) | HR 0.99 (0.90 to 1.08) | 3.5 | 0 | No change | No |
| 9 | 32 (6812) | 8 (3831) | 5 (2942) | VPI total score (U) | SMD -0.21 (-0.29 to -0.12) | SMD -0.19 (-0.28 to -0.11) | 9.0 | 10 | Decrease efficacy | No |
| 7 | 9 (2857) | 1 (514) | 1 (514) | UPDRS scale (U) | MD -1.77 (-2.13 to -1.41) | MD -1.66 (-1.99 to -1.32) | 16.1 | 9 | Decrease efficacy | No |
| 00 | 23 (18 980) | 28 (14 733) | 21 (11 298) | HbA1c (U) | MD -0.35 (-0.51 to -0.19) | MD -0.45 (-0.55 to -0.36) | 58.3 | 29 | Increase efficacy | No |
| Harms: | | | | | | | | | | |
| 6 | 14 (42 602) | 1 (166) | 1 (166) | Major bleeding (U) | OR 0.88 (0.79 to 0.99) | OR 0.88 (0.79 to 0.98) | 0.2 | 0 | No change | No |
| 10 | 70 (32 054) | 4 (2039) | 4 (2039) | Opportunistic infection (PO) | OR 1.79 (1.17 to 2.74) | OR 1.52 (1.04 to 2.23) | 18.7 | 28 | Less harm | No |
| 11 | 9 (11 007) | 4 (810) | 2 (550) | Withdrawal due to adverse event (SO) | RR 0.83 (0.74 to 0.93) | RR 0.85 (0.76 to 0.94) | 0.2 | 13 | More harm | No |
| 12 | 16 (33 958) | 1 (129) | 1 (129) | Major bleeding (PO) | RR 0.79 (0.52 to 1.19) | RR 0.80 (0.54 to 1.20) | 1.5 | 5 | More harm | No |
| 13 | 19 (101 801) | 2 (317) | 2 (317) | Treatment discontinuation due to all cause (PO) | RR 1.40 (1.08 to 1.82) | RR 1.37 (1.06 to 1.75) | 8.6 | 6 | Less harm | No |
| 14 | 43 (16 011) | 7 (943) | 2 (477) | ratal adverse event (U) | RR 1.63 (1.32 to 2.01) | RR 1.62 (1.32 to 2.99) | 1.2 | + | Less harm | No |
| ID=identifi risk ratio (f Rating Scal | cation of review (RR) and odds ratic le; MD=mean diff | see appendix 4, 0 (OR) relates to erence; HbA1 _c =€ |); SR=systematic reviev log values; PASI 75=75 slycated haemoglobin | v; MA=meta-analysis; PO=outcome defined as prir % reduction in Psoriasis Area and Severity Index; | mary in SR; SO=outcome defined HR=hazard ratio; NPI=neuropsy | as secondary in SR; U=primary :hiatric inventory; SMD=standar | and secondary out dised mean differer | come not pres ice; UPDRS sc | pecified in SR; percen ale=Unified Parkinso | tage chang n's Disease |

included only one meta-analysis for each systematic review. Finally, we focused on only systematic reviews of pharmaceutical treatments and cannot extrapolate to non-pharmaceutical treatments because the regulation for trial registration and posting of results is less stringent with these treatments.

Implications for clinicians and policy makers

Clinical trial registries have been developed to reduce waste in research, and publication bias. Their use has been enforced by editors and policy makers. They have been considered to contribute to greater transparency and increasing the value of research.

Searching clinical trial registries is recommended when performing systematic reviews. In our study, the addition of new RCTs in meta-analyses affected treatment effect estimates but did not change the statistical significance of the results or the direction of the treatment effect, although it increased precision.

Nevertheless, searching clinical trial registries remains an essential recommendation for the conduct of systematic reviews and should be enforced. In fact, the objective of systematic reviews is to collate all empirical evidence.³ However, overall, results for only about half of clinical trials are published, and searching only electronic bibliographic databases gives access to just the "tip of the iceberg".²⁴⁻²⁶ Finally, searching trial registries in general represented a low burden. The number of records to screen by systematic review was low: median 23 (interquartile range 6-150). The results for 41 of 63 trials were posted at ClinicalTrials.gov and therefore were immediately available. Furthermore, a previous study showed that the reporting of results was more complete at ClinicalTrials.gov than in published reports.34 However, one important limitation of this search is the lack of availability of the results for completed trials and the low level of details on the methodological quality recorded in the registries. Some initiatives to facilitate access to clinical trial results, such as the Food and Drug Administration Amendments Act 2007, which requires the posting of clinical trial results18 or pharmaceutical company policies,35 have been implemented. Some researchers have developed an intervention to improve posting, such as emailing a reminder about section 801 of the Food and Drug Administration Amendments Act requirement to responsible parties³⁶; other interventions are necessary. Recently, the new rules of trial registration at ClinicalTrials.gov require submission of a full protocol and statistical analysis plan at the same time as submission of results.37 Registries could be an even more important source of results in the future.

Conclusion

Searching clinical trial registries is essential for identifying additional trials that could increase the value of systematic reviews. However, the lack of available RCT results limits the value of the search. Searches of trial registries should be promoted and enforced, as should the posting of trial results.

We thank Laura Smales (BioMedEditing, Toronto, Canada) for English language proofreading and Carolina Riveros for data extraction.

Contributors: MB, AY, IB, and PR conceived, designed the study, and wrote the protocol MB, IB, and AY wrote the first draft. AY, IB, EP, and PR contributed to the writing of the manuscript. MB, AY, EP, and IB analysed the data. AY and IB had access to all of the data in the study and take responsibility for the integrity of the data and the accuracy of the data analysis. MB and AY contributed equally to the manuscript. AY is the guarantor.

Funding: This study received no funding.

Competing interests: All authors have completed the ICMJE uniform disclosure form at www.icmje.org/coi_disclosure.pdf and declare: no support from any organisation for the submitted work; no financial relationships with any organisations that might have an interest in the submitted work in the previous three years; no other relationships or activities that could appear to have influenced the submitted work.

Ethical approval: Not required.

Data sharing: All data from this study—including literature searches, additional explanatory material, and data extraction forms—are available on request.

Transparency: The lead author (AY) affirms that the manuscript is an honest, accurate, and transparent account of the study being reported; that no important aspects of the study have been omitted; and that any discrepancies from the study as planned have been explained.

This is an Open Access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: http://creativecommons.org/licenses/ by-nc/4.0/.

- Cook DJ, Mulrow CD, Haynes RB. Systematic reviews: synthesis of best evidence for clinical decisions. Ann Intern Med 1997;126:376-80. doi:10.7326/0003-4819-126-5-199703010-00006.
- 2 Bastian H, Glasziou P, Chalmers I. Seventy-five trials and eleven systematic reviews a day: how will we ever keep up?PLoS Med 2010;7:e1000326. doi:10.1371/journal.pmed.1000326.
- 3 Higgins JPT, Green S. Cochrane handbook for systematic reviews of interventions. *Cochrane Collab* 2011;version 5. http://handbook. cochrane.org/
- 4 Higgins JP, Altman DG, Gøtzsche PC, et al. Cochrane Bias Methods Group Cochrane Statistical Methods Group. The Cochrane Collaboration's tool for assessing risk of bias in randomised trials. *BMJ* 2011;343:d5928. doi:10.1136/bmj.d5928.
- 5 Moher D, Tetzlaff J, Tricco AC, Sampson M, Altman DG. Epidemiology and reporting characteristics of systematic reviews. *PLoS Med* 2007;4:e78. doi:10.1371/journal.pmed.0040078.
- 6 Ebrahim S, Bance S, Athale A, Malachowski C, Ioannidis JP. Meta-analyses with industry involvement are massively published and report no caveats for antidepressants. *J Clin Epidemiol* 2016;70:155-63. doi:10.1016/j.jclinepi.2015.08.021.
- 7 Chalmers I, Glasziou P. Avoidable waste in the production and reporting of research evidence. *Lancet* 2009;374:86-9. doi:10.1016/ S0140-6736(09)60329-9.
- 8 Chan AW, Song F, Vickers A, et al. Increasing value and reducing waste: addressing inaccessible research. *Lancet* 2014;383:257-66. doi:10.1016/S0140-6736(13)62296-5.
- 9 Easterbrook PJ, Berlin JA, Gopalan R, Matthews DR. Publication bias in clinical research. *Lancet* 1991;337:867-72. doi:10.1016/0140-6736(91)90201-Y.
- 10 Song F, Parekh S, Hooper L, et al. Dissemination and publication of research findings: an updated review of related biases. *Health Technol Assess* 2010;14:iii, ix-xi, 1-193. doi:10.3310/hta14080.
- 11 Hopewell S, Loudon K, Clarke MJ, Oxman AD, Dickersin K. Publication bias in clinical trials due to statistical significance or direction of trial results. *Cochrane Database Syst Rev* 2009;MR000006:MR000006. doi:10.1002/14651858.MR000006.pub3.
- 12 Rising K, Bacchetti P, Bero L. Reporting bias in drug trials submitted to the Food and Drug Administration: review of publication and presentation. *PLoS Med* 2008;5:e217, discussion e217. doi:10.1371/ journal.pmed.0050217.
- 13 Dwan K, Gamble C, Williamson PR, Kirkham JJ. Reporting Bias Group. Systematic review of the empirical evidence of study publication bias and outcome reporting bias - an updated review. *PLoS One* 2013;8:e66844. doi:10.1371/journal.pone.0066844.
- 14 Schmucker C, Schell LK, Portalupi S, et al. OPEN consortium. Extent of non-publication in cohorts of studies approved by research ethics committees or included in trial registries. *PLoS One* 2014;9:e114023. doi:10.1371/journal.pone.0114023.
- 15 Jefferson T, Jones MA, Doshi P, et al. Neuraminidase inhibitors for preventing and treating influenza in healthy adults and children. *Cochrane Database Syst Rev* 2014;CD008965:CD008965. doi:10.1002/14651858.CD008965.pub4.

- 16 Jefferson T, Jones M, Doshi P, Del Mar C. Neuraminidase inhibitors for preventing and treating influenza in healthy adults: systematic review and meta-analysis. *BMJ* 2009;339:b5106. www.ncbi.nlm.nih.gov/ pubmed/19995812. doi:10.1136/bmj.b5106.
- 17 De Angelis C, Drazen JM, Frizelle FA, et al. International Committee of Medical Journal Editors. Clinical trial registration: a statement from the International Committee of Medical Journal Editors. Ann Intern Med 2004;141:477-8. doi:10.7326/0003-4819-141-6-200409210-00109.
- 18 United States Congress. (2007) Food and Drug Administration Amendments Act (FDAAA) of 2007: public law no 110-85. www.gpo. gov/fdsys/pkg/PLAW-110publ85/pdf/PLAW-110publ85.pdf
- Groves T, Mandatory disclosure of trial results for drugs and devices. BMJ 2008;336:170. doi:10.1136/bmj.39469.465139.80.
- 20 Laine C, Horton R, DeAngelis CD, et al. Clinical trial registration: looking back and moving ahead. JAMA 2007;298:93-4. doi:10.1001/ jama.298.1.jed70037.
- 21 Shea BJ, Grimshaw JM, Wells GA, et al. Development of AMSTAR: a measurement tool to assess the methodological quality of systematic reviews. BMC Med Res Methodol 2007;7:10. doi:10.1186/1471-2288-7-10.
- 22 Liberati A, Altman DG, Tetzlaff J, et al. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration. J Clin Epidemiol 2009;62:e1-34. doi:10.1016/j.jclinepi.2009.06.006.
- 23 Moher D, Liberati A, Tetzlaff J, Altman DG. PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *BMJ* 2009;339:b2535. doi:10.1136/bmj.b2535.
- 24 Keil LG, Platts-Mills TF, Jones CW. Systematic reviews published in emergency medicine journals do not routinely search clinical trials registries: a cross-sectional analysis. *Ann Emerg Med* 2015;66:424-7. doi:10.1016/j.annemergmed.2014.10.001.
- 25 van Enst WA, Scholten RJ, Hooft L. Identification of additional trials in prospective trial registers for Cochrane systematic reviews. *PLoS One* 2012;7:e42812. doi:10.1371/journal.pone.0042812.
- 26 Jones CW, Keil LG, Weaver MA, Platts-Mills TF. Clinical trials registries are under-utilized in the conduct of systematic reviews: a cross-sectional analysis. Syst Rev 2014;3:126. doi:10.1186/2046-4053-3-126.
- 27 Cortés J, González JA, Campbell MJ, Cobo E. A hazard ratio was estimated by a ratio of median survival times, but with considerable uncertainty. *J Clin Epidemiol* 2014;67:1172-7. doi:10.1016/j.jclinepi.2014.05.017.
- 28 Hart B, Lundh A, Bero L. Effect of reporting bias on meta-analyses of drug trials: reanalysis of meta-analyses. *BMJ* 2012;344:d7202. doi:10.1136/bmj.d7202.
- 29 Dechartres A, Boutron I, Trinquart L, Charles P, Ravaud P. Single-center trials show larger treatment effects than multicenter trials: evidence from a meta-epidemiologic study. Ann Intern Med 2011;155:39-51. doi:10.7326/0003-4819-155-1-201107050-00006.
- 30 Potthast R, Vervölgyi V, McGauran N, Kerekes MF, Wieseler B, Kaiser T. Impact of inclusion of industry trial results registries as an information source for systematic reviews. *PLoS One* 2014;9:e92067. doi:10.1371/ journal.pone.0092067.
- 31 Page MJ, Shamseer L, Altman DG, et al. Epidemiology and reporting characteristics of systematic reviews of biomedical research: a cross-sectional study. *PLoS Med* 2016;13:e1002028. doi:10.1371/ journal.pmed.1002028.
- 32 Golder S, Loke YK, Wright K, Norman G. Reporting of adverse events in published and unpublished studies of health care interventions: a systematic review. *PLoS Med* 2016;13:e1002127. doi:10.1371/journal. pmed.1002127.
- 33 Glanville JM, Duffy S, McCool R, Varley D. Searching ClinicalTrials.gov and the International Clinical Trials Registry Platform to inform systematic reviews: what are the optimal search approaches? J Med Libr Assoc 2014;102:177-83. doi:10.3163/1536-5050.102.3.007.
- 34 Riveros C, Dechartres A, Perrodeau E, Haneef R, Boutron I, Ravaud P. Timing and completeness of trial results posted at ClinicalTrials.gov and published in journals. *PLoS Med* 2013;10:e1001566, discussion e1001566. doi:10.1371/journal.pmed.1001566.
- 35 Strom BL, Buyse M, Hughes J, Knoppers BM. Data sharing, year 1--access to data from industry-sponsored clinical trials. N Engl J Med 2014;371:2052-4. doi:10.1056/NEJMp1411794.
- 36 Maruani A, Boutron I, Baron G, Ravaud P. Impact of sending email reminders of the legal requirement for posting results on ClinicalTrials. gov: cohort embedded pragmatic randomized controlled trial. *BMJ* 2014;349:g5579. doi:10.1136/bmj.g5579.
- 37 Zarin DA, Tse T, Williams RJ, Carr S. Trial reporting in ClinicalTrials. gov—the final rule. N Engl J Med 2016;375:1998-2004. doi:10.1056/ NEJMsr1611785.

Appendices 1-3: Appendix 1, verification of recruitment status according to registry; appendix 2, keywords, date of search, and detection in WHO ICTRP search portal; and appendix 3, systematic reviews for which data could not be added to meta-analysis **Appendix 4:** Impact of trial registry searches on summary statistics